Author's response to reviews

Title: Functional polymorphism of the NFKB1 gene promoter is related to the risk of dilated cardiomyopathy

Authors:

Bin Zhou (zb630@163.com)
Li Rao (lirao@mail.sc.cninfo.net)
Ying Peng (joeypengying@yahoo.com.cn)
Yanyun Wang (wyanyun@gmail.com)
Yi Li (lizzy619@163.com)
Linbo Gao (uglyduckling726@163.com)
Yu Chen (polyb@163.com)
Hui Xue (xuehu1212@163.com)
Yaping Song (songyaping12@163.com)
Miao Liao (liaomiao@scu.edu.cn)
Lin Zhang (zhanglin@scu.edu.cn)

Version: 4 Date: 22 April 2009

Author's response to reviews: see over
Dear Editor:

Thanks very much for your attention to our paper and we are really sorry for the delay in the revision process.

We have revised our manuscript in a very careful way according to the editorial board members comments. The revised manuscript was entitled “R2--Functional polymorphism of the NFKB1 gene promoter is related to the risk of dilated cardiomyopathy”.

According to your comments, we improved the discussion and conclusions section, which are marked in red.

We look forward to receiving your decision and we hope that our manuscript is suited as an original article in your journal BMC Medical Genetics.

Lin Zhang, Ph.D.

Laboratory of Molecular Translational Medicine

West China Second University Hospital

Sichuan University

Chengdu, 610041, Sichuan, P. R. China

Tel: +86 (028) 85469033

Fax: +86 (028) 85401825

Email: zhanglin@scu.edu.cn

Sincerely yours,

Lin Zhang Ph.D.